Dear Editor,

Thank you for considering our manuscript entitled “Understanding test accuracy research: a test consequence graphic” for publication in Diagnostic and Prognostic Research. We are very pleased that you are interested in publishing our paper.

We would like to thank the reviewers for their positive and helpful comments. We have expanded the methods section to explain more about the process used to develop the graphic. We have also added a web appendix to show the different versions of the graphic as it evolved through the process.
We hope that you will now consider our manuscript suitable for publication,

With best wishes

Dr Penny Whiting